Author’s response to reviews

Title: Factors associated with myopia in Korean children: Korea National Health and Nutrition Examination Survey 2016–2017 (KNHANES VII)

Authors:

Hyuna Kim (landyonel@gmail.com)
Jin Seo (jinseok90@naver.com)
Woong-Sun Yoo (oocee@daum.net)
Gyu-Nam Kim (sayuh@naver.com)
Rock Kim (krb747@gmail.com)
Jae Chae (chaejaeun@gmail.com)
Inyoung Chung (inyoung@gnu.ac.kr)
Seong-Wook Seo (stramast@naver.com)
Seong-Jae Kim (oocee@naver.com)

Version: 3 Date: 08 Nov 2019

Author’s response to reviews:

November 8th, 2019

Nabin Paudel, Ph.D
Editor
BMC Ophthalmology


Dear Paudel,

We checked your careful comments and answered as point-by-point response.

Please find attached revised manuscript and figures.
1. Ethics

In the "Ethics and consent to participate" section of your Declarations, please clarify if the approval stated was from an ethics committee or institutional review board from the Korean Center for Disease Control.

; The “institutional review board of the Korean center for disease control” was added into appropriate part.

2. Remove supplementary file

Please remove the STROBE guidelines checklist from the manuscript, as these files are not required to be attached to the manuscript at this stage of the editorial process.

; Removed as commented.

3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

; We checked the cleared up manuscript as final version.

We deeply appreciate your time and effort for our manuscript.

Sincerely,

Seong-Jae, Kim

Department of Ophthalmology, Gyeongsang National University Hospital and Gyeongsang National University college of medicine, 79 Gangnam-ro, Jinju, Gyeongnam 660-702 South Korea
E-mail: maya12kim@naver.com
Tel: +82-55-750-8468
Fax: +82-55-758-4158